Author's response to reviews

Title: TECNOB: study design of a randomized controlled trial of a multidisciplinary telecare intervention for obese patients with type-2 diabetes.

Authors:

Gianluca Castelnuovo (gianluca.castelnuovo@auxologico.it)
Gian Mauro Manzoni (gm.manzoni@auxologico.it)
Paola Cuzziol (p.cuzziol@auxologico.it)
Gian Luca Cesa (gluca.cesa@gmail.com)
Cristina Tuzzi (c.tuzzi@auxologico.it)
Valentina Villa (v.villa@auxologico.it)
Antonio Liuzzi (liuzzi@auxologico.it)
Maria Letizia Petroni (ml.petroni@auxologico.it)
Enrico Molinari (molinari@auxologico.it)

Version: 3 Date: 23 February 2010

Author's response to reviews:

Dear Editor,

sorry for the missing.

I revised the Competing Interest section. I clarified that part of the technonological devices used in the study are commercially available, except for the web-platform, and that authors held on no financial interest therein.

Sincerely,

Gianluca Castelnuovo